Author's response to reviews

Title: The association of psychiatric comorbidity and use of the emergency department among persons with substance use disorders: an observational cohort study

Authors:

Geoffrey M Curran (currangeoffreym@uams.edu)
Greer Sullivan (sullivangreer@uams.edu)
Keith Williams (williamsdavidk@uams.edu)
Xiaotong Han (Xiaotonghan@uams.edu)
Elise Allee (alleemarye@uams.edu)
Kathryn J Kotrla (kotrla@medicine.tamhsc.edu)

Version: 4 Date: 19 November 2008

Author's response to reviews:

The title and tables have been reformatted. The abstract matches what is in the submission system.